Author's response to reviews

Title: A case of squamous cell carcinoma arising from the suprapubic cystostomy tract

Authors:

Hiroki Ito (pug_daikichi@yahoo.co.jp)
Masayuki Arao (tanntann.com@gmail.com)
Hanako Ishigaki (hanako.9.11@hotmail.co.jp)
Noboru Ohshima (noboruswk@gmail.com)
Ayako Horita (mainichi-daikichi@ezweb.ne.jp)
Ikuo Saito (sleepynaoki@yahoo.co.jp)
Kotaro Hirai (k-hirai@sagamihara-hosp.gr.jp)

Version: 2 Date: 17 June 2011

Author's response to reviews: see over
Dear Editors:

Enclosed please find our manuscript entitled, “A case of squamous cell carcinoma arising from the suprapubic cystostomy tract”, which I am submitting for publication in BMC urology. This case presentation is a forth report of squamous cell carcinoma arising the suprapubic catheter tract in the previous literatures. I think that BMC urology is one of the most reliable journal particularly for the case presentation.

This manuscript does not contain any data that have been published or been submitted for publication elsewhere, and its contents have not been presented in whole or in part at society meetings, etc. I attest to the fact that all of the authors have read the manuscript, have attested to the validity and legitimacy of the data and its interpretation, and have agreed to its submission to BMC urology for publication. We have no conflicts of interest.

I hope that you find our manuscript suitable for publication in BMC urology.

We made background and conclusion, consent sections.

Thank you for your revision.

Sincerely yours,
Hiroki Ito